Author’s response to reviews

Title: Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study

Authors:

Nobuyuki Nishikawa (nishikaw@kuhp.kyoto-u.ac.jp)
Rie Yago (yago@kc.twmu.ac.jp)
Yuichiro Yamazaki (yuichiro@gmail.com)
Hiromitsu Negoro (hnegoro@kuhp.kyoto-u.ac.jp)
Mari Suzuki (msuzuki@kc.twmu.ac.jp)
Masaaki Imamura (m69ima@kuhp.kyoto-u.ac.jp)
Yoshinobu Toda (toda@tenriyorozu-u.ac.jp)
Kazunari Tanabe (tanabe@kc.twmu.ac.jp)
Osamu Ogawa (ogawao@kuhp.kyoto-u.ac.jp)
Akihiro Kanematsu (aquil@hyo-med.ac.jp)

Version: 4 Date: 23 December 2014

Author’s response to reviews: see over
Ms Hazel Joyce Delos Santos  
Journal Editorial Office  
BMC Urology  

December the 18th, 2014  

MS ID.: 5131817871403085  
Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study  

Dear Ms Hazel Joyce Delos Santos  
Journal Editorial Office  

Please find attached a revised version of our revised manuscript entitled “Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study.”  

We have addressed the comments by The BioMed Central Editorial Team, and have revised the manuscript in this version, as shown in the following pages.  

We hope that in its current form the manuscript is suitable for peer review in BMC Urology.  

Thank you again for your help and interest in our work.  

Yours sincerely,  

Akihiro Kanematsu MD, PhD  
Associate Professor  
Department of Urology, Hyogo College of Medicine  
1-1 Mukogawa-cho, Nishinomiya, Hyogo,  
663-8501 Japan
We have revised our manuscript to resolve the problem raised by the editor in the previous version. Our answers to the editors’ comments are as follows:
The change to the revised manuscript is marked in yellow.

(1) Tables
Tables were included in the main manuscript, and removed the files from the additional files.

Table2:
The right subtitle of the column under Maximal bladder capacity was changed due to a grammatical mistake.

Percent capacity of age standard